



# Article Multiphysics Modeling of Power Transmission Line Failures Across Four US States

Prakash KC<sup>1</sup>, Maryam Naghibolhosseini<sup>2</sup> and Mohsen Zayernouri<sup>1,3,\*</sup>

- <sup>1</sup> Department of Mechanical Engineering, Michigan State University, East Lansing, MI 48824, USA; kprakash@msu.edu
- <sup>2</sup> Department of Communicative Sciences and Disorders, Michigan State University, East Lansing, MI 48824, USA; naghib@msu.edu
- <sup>3</sup> Department of Statistics and Probability, Michigan State University, East Lansing, MI 48824, USA
- \* Correspondence: zayern@msu.edu

**Abstract:** The failure of overhead transmission lines in the United States can lead to significant economic losses and widespread blackouts, affecting the lives of millions. This study focuses on analyzing the failure of transmission lines, specifically considering the effects of wind, ambient temperature, and current demands, incorporating minimal and significant pre-existing damage. We propose a multiphysics framework to analyze the transmission line failures across sensitive and affected states of the United States, integrating historical data on wind and ambient temperature. By combining numerical simulation with historical data analysis, our research assesses the impact of varying environmental conditions on the reliability of transmission lines. Our methodology begins with a deterministic approach to model temperature and damage evolution, using phase-field modeling for fatigue and damage coupled with electrical and thermal models. Later, we adopt the probability collocation method to investigate the stochastic behavior of the system, enhancing our understanding of uncertainties in model parameters, conducting sensitivity analysis to identify the most significant model parameters, and estimating the probability of failures over time. This approach allows for a comprehensive analysis of factors affecting transmission line reliability, contributing valuable insights into improving power line's resilience against environmental conditions.

**Keywords:** transmission line; finite element method; probability collocation method; uncertainty quantification; sensitivity analysis; probability of failure

## 1. Introduction

The power grid's components are so interconnected that the failure of a single component can cause a widespread outage. Overhead transmission lines exposed to dynamic weather conditions and current loads are particularly vulnerable. Such vulnerabilities not only lead to frequent blackouts across the United States, affecting millions annually, but also disrupt daily life and impose significant economic burdens. A report by the Department of Energy estimated that weather-related power outages cost the US economy between USD 18 billion and USD 33 billion from 2003 to 2012 [1]. Some states such as California, Texas, and Michigan are particularly more vulnerable, with Florida having the highest total number of 25,348,824 affected by power outages [2]. Hence, understanding the reliability of transmission cables in such sensitive and affected regions is crucial.

Several studies have investigated the impact of environmental conditions on overhead power lines. The effect on the life of aluminum conductor steel-reinforced (ACSR) cables due to thermal stress was investigated by integrating a strength reliability analysis based on Monte Carlo simulation and fuzzy logic-based ductility analysis [3]. Additionally, the damage caused by wind loading has been extensively studied, affecting the transmission lines and the supporting towers [4]. Studies by [5–7] have extensively examined the dynamic responses and structural effects of wind on transmission line infrastructure. In [8],



Citation: KC, P.; Naghibolhosseini, M.; Zayernouri, M. Multiphysics Modeling of Power Transmission Line Failures Across Four US States. *Modelling* 2024, *5*, 1745–1772. https://doi.org/10.3390/ modelling5040091

Received: 7 October 2024 Revised: 8 November 2024 Accepted: 14 November 2024 Published: 20 November 2024



**Copyright:** © 2024 by the authors. Licensee MDPI, Basel, Switzerland. This article is an open access article distributed under the terms and conditions of the Creative Commons Attribution (CC BY) license (https:// creativecommons.org/licenses/by/ 4.0/). the overheating effect of wildfire due to the radiative heat transfer was studied. Similarly, the ice and wind combined effect was studied in [9]. Several studies have been made to understand the effect of weather conditions on the thermal rating of the conductor [10-12]. In addition to environmental factors, the choice of conductor material plays a crucial role in determining the reliability of transmission wires.

All these studies laid a solid basis for understanding the failure mechanisms of power transmission. While much of this research has focused on the thermal and electrical modeling of transmission lines, none considered the mechanical model that relates to fatigue and damage that influence the overall conductivity of the material. Transmission lines are subjected to dynamic wind, temperature, and electrical loads, which can induce fatigue and damage to the wires. This damage compromises the conductivity of the conductor, leading to increased heat generation. Therefore, a coupled model that integrates mechanical properties with electrical and thermal aspects can significantly improve failure prediction.

Recently, phase-field models have been integrated with electrical and thermal models to study various phenomena [13–15]. Over the years, these models have been crucial in simulating brittle [16–21], ductile [22–26], isothermal fatigue fracture [27], and non-isothermal fatigue fracture mechanisms [28], They accurately capture phenomena such as crack initiation, propagation, branching, and coalescence—events commonly observed in dynamic fractures. Based on these applications, incorporating the phase-field model with electrical and thermal models becomes crucial in studying transmission line reliability. Further incorporation of stochastic analysis can enhance the precision of lifespan predictions. This approach enables a more comprehensive understanding of the factors that affect transmission line durability in real-world scenarios.

Although the phase-field methods are extensively used for damage evolution, discrete dislocation dynamics (DDD) can provide valuable insights into the underlying microstructural mechanisms [29]. The interaction of dislocation can significantly affect the behavior resulting in failure [30,31]. The normal behavior of materials also changes under high operating temperatures, resulting in visco-elastic behavior. Integrating fractional visco-elasto-plastic models as in the studies [32–35] can significantly improve the reliability analysis. Even considering the uncertainty in the fractional order further improves capturing the behavior of the system [36].

The classical Monte Carlo (MC) method [37,38] is a standard benchmark for studying stochastic solutions that compute the quantity of interest (QoI) in a straightforward fashion. However, this method is characterized by a slow convergence rate and requires a large number of realizations. Some well-established methods, such as polynomial chaos [39,40] or its generalization via Galerkin projection [41–43], require modifications to the governing equations for stochastic analysis. This requirement makes these methods intrusive and potentially impractical for complex problems since such modifications may not be feasible or could overly complicate the analysis. To address this, non-intrusive techniques must be used; one such technique is the probability collocation method (PCM) [44,45]. PCM preserves the simplicity of solution structures and allows for independently sampled realizations, thereby achieving better convergence compared to the traditional Monte Carlo (MC) method. Although the curse of dimensionality challenges PCM due to tensorial products, this issue can be effectively addressed using techniques such as sparse grids [46] or active subspace methods [47–49].

In a recent study [50], the authors integrated the phase-field model with thermal and electrical modules to predict the lifespan of overhead transmission lines. The study initially examined a deterministic solution and later implemented the PCM for stochastic analysis. However, a more comprehensive analysis involving historical data across various specific states would provide a more realistic assessment. The reliability and durability of transmission lines vary under different weather conditions, highlighting the need to adjust the weather condition assessments to particular cases. Furthermore, analyzing the lifespan of transmission lines with insignificant initial damage could offer valuable insights into the extent to which pre-existing damage influences overall reliability.

In this study, we develop a coupled fatigue and damage-based phase-field model with thermal and electrical aspects to investigate the reliability of overhead transmission lines under environmental conditions and the presence of initial damage, including the uncertainties associated with these. We first investigate the reliability of Transmission Lines deterministically using the finite element method. To explore the uncertainties associated with environmental conditions and initial damage, we incorporate PCM for uncertainty quantification (UQ), sensitivity analysis (SA), and probability of failure using the limit state function. To understand the reliability of transmission lines more realistically, we incorporate several realistic scenarios for a detailed historical analysis of wind and temperature data rather than using parameterized data by focusing on the four most affected U.S. states: Texas (TX), California (CA), Michigan (MI), and Florida (FL) according to [2]. Additionally, we assess the system's reliability under minimal damage to establish a baseline and examine how precursor damage affects the reliability of transmission lines. By analyzing these scenarios using our developed model, this research offers critical insights into the conditions-including pre-existing defects-that most significantly influence the durability and reliability of transmission lines. These insights can guide the development of more resilient power line infrastructures designed to withstand diverse environmental conditions and extend their lifespan. Ultimately, this work contributes to improving the reliability of the transmission line.

This paper is structured as follows: Section 2 presents the problem statement, including four representative scenarios. Section 3 presents a multiphysics model and discusses each model. Additionally, the section discusses the historical data analysis for wind and temperature of each specific scenario, including the reliability analysis procedure using a limit state function. Section 4 presents the deterministic approach and solution, including one-dimensional finite element discretization. Section 5 presents the stochastic analysis methods and solutions using PCM as the building block. Section 6 presents the verification and analysis of the model. Section 7 addresses the conclusion.

# 2. Problem Statement

The operating temperature is crucial for the reliable operation of overhead transmission lines. High operating temperature affects the wire's life and reliability. Even the presence of damage in the wire further elevates the temperature These issues often remain undetected until a cable rupture occurs. Additionally, environmental conditions, including temperature, speed of wind, and current load, can adversely affect these parameters, pushing them beyond acceptable levels.

Therefore, operating temperature is considered the primary factor for our analysis. Along with physical effects, material parameters, and loading conditions, these elements determine the failure state of the transmission line over its lifespan. This study aims to understand the multiphysics effects on these critical factors, which are crucial for assessing the transmission line's reliability.

## Representative Scenarios

Previous research on the reliability of transmission lines often relied on parameterized values for wind and temperature, which lack precision. This study aims to improve this by incorporating real data from four representative scenarios: TX, CA, MI, and FL. We source wind and temperature data from [51,52]. The data for each state are presented in Table 1.

TX -	ft/s	11.59	12.32	12.76	13.05	12.17	10.41	9.09	8.65	9.24	10.12	10.85	11.15
	Κ	287.54	286.98	294.82	294.71	298.71	301.21	303.48	303.48	300.37	295.21	292.04	286.15
CA -	ft/s	2.05	2.79	3.08	3.52	3.23	3.08	3.08	2.93	2.49	2.20	2.05	2.05
	Κ	286.65	288.04	287.76	290.09	293.37	294.21	295.71	297.54	297.15	295.15	289.76	287.65
MI	ft/s	16.13	15.40	15.25	15.25	13.35	12.32	11.59	10.71	11.29	13.49	15.11	15.40
	Κ	273.37	272.09	278.26	280.98	287.43	294.87	298.43	296.21	290.59	283.71	280.76	273.65
FL -	ft/s	12.61	12.91	14.23	14.23	13.20	11.00	10.56	10.41	10.85	12.61	12.91	12.17
	К	295.09	295.98	298.26	300.87	299.48	302.26	303.09	302.76	301.93	301.04	298.43	293.93

Table 1. Wind speed (ft/s) [51] and temperature (K) [52] data by state.

## 3. Methodology

This study aims to analyze the failure mechanisms in overhead transmission lines by developing a coupled model that integrates thermal, mechanical, and electrical aspects. We systematically assess how mechanical properties, in terms of damage, affect the electrical and thermal aspects of the cable in the presence of environmental conditions contributing to line reliability. Figure 1 illustrates the transmission line mounted on the transmission tower, highlighting the cable failure.



Figure 1. Schematic representation of transmission lines.

We consider a cable to be a one-dimensional domain. The cable is supported by two towers to maintain the sag on it. We consider the projected span equal to its effective operational length for simplification, as shown in Figure 2. While sag can be an essential consideration, we only account for its influence on horizontal tension due to temperature variations. We do not explicitly model the sag; instead, we focus on understanding how the horizontal tension contributes to material damage and fatigue over time.



Figure 2. One-dimensional representation of transmission line.

The operating temperature of the conductor affects the tension acting on it. High temperature reduces the tension, while low temperature increases it due to the effect of sag. The temperature of the cable is mainly driven by current passing through it and convective cooling from the wind acting on it. Additionally, any damage to the cable reduces its conductivity, resulting in further heat generation.

The electrical model here does not consider the electrical design of transmission lines. Rather, it focuses on the effect of current passing through the conductor independent of the voltage levels delivered to consumers.

In this study, we propose that cyclic loading accelerates the damage to the material and reduces the conductivity. This further increases the temperature due to resistivity effects and leads to greater heat generation due to Joule heating, ultimately leading to failure. The heat generation elevates the temperature of the material and accelerates aging.

This section introduces the mechanical, thermal, and electrical models that collectively simulate these processes. We detail the method for deriving cyclic loading using DFT and assessing transmission line reliability under these conditions.

#### 3.1. Mechanical Model

For the mechanical model, we followed the principles in [28], which provides the governing equations for displacement  $u_d$ , damage  $\varphi_d$ , and fatigue  $\mathcal{F}_f$ .  $\varphi_d = 0$  represents virgin material, and  $\varphi_d = 1$  indicates complete damage, while values  $0 < \varphi_d < 1$  represent intermediate levels of damage.  $\mathcal{F}_f$  is considered an internal variable. Specific material evolution behaviors can be obtained from the governing equations by choosing appropriate free-energy potentials. Alternatively, one can approach this by considering the free-energy function, as follows:

$$\Psi(\nabla u_d, \varphi_d, \nabla \varphi_d, \mathcal{F}_f) = d(\varphi_d) Y(\nabla u_d)^2 + g_c \frac{\gamma}{2} (\nabla \varphi_d)^2 + \mathcal{K}(\varphi_d, \mathcal{F}_f), \tag{1}$$

where *Y* (GPa) denotes the Young's modulus,  $g_c$  (kN/m) denotes the fracture energy release rate, and  $\gamma > 0$  denotes the phase-field layer width parameter (m). The degradation function is defined as  $d(\varphi_d) = (1 - \varphi_d)^2$ , affecting both elastic response and the electrical conductivity. Additionally,  $\mathcal{K}(\varphi_d, \mathcal{F}_f)$  represents the coupling of damage and fatigue, describing how damage evolves due to fatigue over time.

The original model described in [28] is time-dependent. However, we considered the quasi-static form for our problem, which allowed us to simplify the governing equations for  $u_d$  and  $\varphi_d$ . Still, the evolution of  $\mathcal{F}_f$ , which represents long-term aging, is modeled as a time-dependent ordinary differential equation (ODE), as follows:

$$\nabla \cdot \left( (1 - \varphi_d)^2 \Upsilon \nabla u_d \right) - \gamma g_c \, \nabla \cdot \, (\nabla \varphi_d \otimes \nabla \varphi_d) + f = 0, \tag{2}$$

$$\gamma g_c \Delta \varphi_d + (1 - \varphi_d) (\nabla u_d)^T Y (\nabla u_d) - \frac{1}{\gamma} [g_c \mathcal{H}'(\varphi_d) + \mathcal{F}_f \mathcal{H}'_f(\varphi_d)] = 0,$$
(3)

$$\dot{\mathcal{F}}_f = -\frac{F_f}{\gamma} \mathcal{H}_f(\varphi_d),\tag{4}$$

The potentials  $\mathcal{H}(\varphi_d)$  and  $\mathcal{H}_f(\varphi_d)$  describe the damage evolution from 0 to 1 as fatigue changes from zero to  $g_c$ .  $\mathcal{H}'(\varphi_d)$  and  $\mathcal{H}'_f(\varphi_d)$  are the derivatives of these potentials with respect to  $\varphi_d$ . The appropriate choices of these potentials are as follows:

$$\mathcal{H}(\varphi_d) = \begin{cases} 0.5\varphi_d^2 & \text{for } 0 \le \varphi_d \le 1, \\ 0.5 + \delta(\varphi_d - 1) & \text{for } \varphi_d > 1, \\ -\delta\varphi_d & \text{for } \varphi_d < 0. \end{cases}$$
(5)

$$\mathcal{H}_{f}(\varphi_{d}) = \begin{cases} -\varphi & \text{for } 0 \leq \varphi_{d} \leq 1, \\ -1 & \text{for } \varphi_{d} > 1, \\ 0 & \text{for } \varphi_{d} < 0. \end{cases}$$
(6)

We describe the evolution of  $\mathcal{F}_f$  through  $\hat{F}_f$ , representing the formation and growth of micro-cracks under cyclic loading conditions and the influence of temperature.  $\hat{F}_f$  has a linear relationship with the level of stress associated with the following virgin material:

$$\hat{F}_f = \rho_m a \left(\frac{\theta_c}{\theta_0}\right) (1 - \varphi_d) |Y \nabla u_d|, \tag{7}$$

where the parameter *a* represents the aging rate,  $\theta_c$  (K) represents the conductor temperature,  $\theta_0$  (K) denotes the reference temperature, and  $\rho_m$  (kg/m<sup>3</sup>) denotes the density of the material.

The mechanical model allows damage healing when tensile stress decreases. To simulate an irreversible damage process and avoid healing mechanisms, we adopt an approach similar to that in [19]. We define  $\mathbb{H}_e$  as the local maximum strain energy history:

$$\mathbb{H}_{e}(x,t) = \max((\nabla u_{d}(x,t))^{T} Y(\nabla u_{d}(x,t)), \mathcal{H}(x,t)).$$
(8)

We integrate the local maximum strain energy history variable  $\mathbb{H}_{e}$  into the damage equation and obtain a modified equation for damage as follows:

$$\gamma g_c \Delta \varphi_d + (1 - \varphi_d) \mathbb{H}_e - \frac{1}{\gamma} [g_c \mathcal{H}'(\varphi_d) + \mathcal{F}_f \mathcal{H}'_f(\varphi_d)] = 0.$$
(9)

#### 3.2. Thermal Model

The original model [28] relates fatigue with the increase in temperature due to repetitive and fast loading; however, here, we focus on the long-term damage rather than the short-term increase in temperature. Although the short-term effects are crucial, the longterm reliability of the transmission lines is mainly affected by static loads, environmental conditions, and gradual material degradation. Therefore, we assume a quasi-static regime to simplify our analysis by focusing on steady-state conditions that significantly impact long-term performance. The steady-state heat equation is given by the following:

$$\nabla \cdot (\kappa_m \nabla \theta_c) + q = 0. \tag{10}$$

Equation (10) describes the heat balance within a system, where  $\kappa_m$  (W/(m K)) represents the thermal conductivity, and q (J/m<sup>2</sup>) denotes the net heat exchange. Joule heating,  $q_j$  (J/m<sup>2</sup>), resulting from the current passing through the conductor, serves as a heat source, while convective cooling,  $q_c$  (J/m<sup>2</sup>), driven by the wind, acts as a heat sink. Thus, the overall heat exchange can be represented as follows:

$$=q_{j}-q_{c}, \tag{11}$$

We consider a convective heat transfer using the following relation:

q

$$q_c = h(\theta_c - \theta_a),\tag{12}$$

where h (W/(m<sup>2</sup> K)) is the convective heat transfer coefficient. We consider forced convection due to cross-flow over the cylinder using the relation given by [53], where h can be calculated using the following relation:

$$Nu_D = \frac{hD}{\kappa_{air}} = C_e R e_D^{m_e} P r^{\frac{1}{3}}$$
(13)

where  $Nu_D$  represents the Nusselt number and Pr represents the Prandtl number. We first determine the Reynolds number  $Re_D$  using a relation given by the following:

$$Re_D = \frac{v_{air}D}{v_{air}},\tag{14}$$

where  $v_{air}$  (m<sup>2</sup>/s) denotes the kinematic viscosity of air,  $v_{air}$  (m/s) denotes the velocity of air, and *D* (m) denotes the diameter of the conductor.

 $C_e$  and  $m_e$  are experimentally determined, and depend upon the value of  $Re_D$ . Table 2 shows the values of  $C_e$  and  $m_e$  for different ranges of  $Re_D$ .

Re <sub>D</sub> Range	$C_e$	$m_e$
0.4–4	0.989	0.330
4-40	0.911	0.385
40-4000	0.683	0.466
4000-40,000	0.193	0.618
40,000-400,000	0.027	0.805

**Table 2.** Experimental values of  $C_e$  and  $m_e$  for different  $Re_D$  ranges.

#### 3.3. Electrical Model

In the transmission line design, various factors such as frequency, inductance, reactance, and electromagnetic interactions with the environment and nearby conductors are considered. However, we are not considering the overall design of the transmission lines; rather, we focus on understanding how current-driven Joule heating affects the reliability of cables with existing damage. Additionally, we do not consider the transient effects of AC currents since they average out over time. Therefore, we use a DC-equivalent mean current, simplifying our approach to measure the heat source term efficiently. Further, to reduce complexity, we parameterize the base value of the current  $I_{base}$ , corresponding to the allowable ampacity for all-aluminum conductors with a diameter of approximately 40 mm. While the current data can be obtained through sophisticated methods, we do not focus on a specific transmission line. Instead, our emphasis is on the voltage drop due to temperature and damage-induced resistivity.

From the above consideration, we focus on solving the conservation of current through the following set of equations:

$$\nabla \cdot J_d = 0, \tag{15}$$

$$J_d = \sigma_E E, \tag{16}$$

$$E = -\nabla V, \tag{17}$$

where  $J_d$  (A/m<sup>2</sup>) denotes current density, *E* (V) denotes the electric field generated by the voltage *V*, and  $\sigma_E$  represents the electric conductivity at the operating temperature, which depends upon the degradation function,  $d(\varphi_d)$ , and non-degraded conductivity of damage,  $\sigma_{E,T}$ :

$$\sigma_E = (1 - \varphi_d)^2 \sigma_{E,T}.$$
(18)

 $\sigma_{E,T}$  at the operating temperature, is related to the conductivity  $\sigma_{E,0}$  at a reference temperature and can be obtained by the following:

$$\sigma_{E,T} = \frac{\sigma_{E,0}}{1 + \alpha_c(\theta_c - \theta_0)},\tag{19}$$

where  $\alpha_c$  is the coefficient of resistivity of the conductor. Combining the above equations, the equation for the voltage field is as follows:

$$\nabla \cdot (-\sigma_E \nabla V) = 0, \tag{20}$$

 $q_i$  can be defined as follows:

$$q_j = J_d \cdot E. \tag{21}$$

Overall, this model links damage to voltage drop, increasing the power loss due to increased resistance, thus aggravating the thermal load due to Joule heating.

#### 3.4. Sag Consideration

The temperature of the cable influences the horizontal tension acting on it. This section focuses on obtaining the mechanical load acting on a cable fixed at two ends, forming a catenary curve. While several studies have been conducted to understand the mechanical behavior of such cables [6,54], we simplify our approach using a one-dimensional damage phase-field model. We assume the cable's length L (m) is approximately equal to the span length  $S_1$  (m). This assumption simplifies the model but does not ignore sag S; rather, it highlights the effect of sag on horizontal tension H that results in fatigue and damage.

The sag in the cable is directly related to the temperature variation affecting the horizontal tension. Higher temperatures increase the length of the cable due to thermal elongation, which increases the sag and reduces tension. Conversely, lower temperatures contract the cable, reducing the length and ultimately reducing the sag and increasing the tension. Therefore, the horizontal tension at either end of the cable results from an initial pre-tension combined with tension adjustments due to temperature variations. Based on these parameters, this model determines the appropriate mechanical loading conditions in terms of horizontal tension.

We follow the methodology presented in [55] to calculate the horizontal tension in the cable. We consider  $W_b$  (kg/m) as the cable weight per unit length and  $H_0$  as the initial tension due to its load, which is considered 20% of the aluminum's ultimate strength. The initial slag  $S_0$  is calculated as follows:

$$S_0 = \frac{W_b S_l^2}{8H_0},$$
 (22)

Although this is a simplification, the theoretical length  $L_0$  needs to be considered, which is given by the following relation:

$$L_0 = S_l + \frac{8S_0^2}{3S_l}.$$
 (23)

The length of the cable changes with the change in the temperature according to the following classical formula:

$$L = L_0 (1 + \alpha_L \Delta \theta_c), \tag{24}$$

where  $\alpha_L$  represents the thermal expansion coefficient. The sag due to the change in length is as follows:

$$S = \sqrt{\frac{3S_l(L-S_l)}{8}},$$
 (25)

which then leads to a new calculation for the horizontal tension as follows:

$$H = \frac{WS_l^2}{8S},\tag{26}$$

In this model, *W* denotes the total weight that could account for the weight of additional factors such as ice and wind. For this study, we consider only the wind component and calculate *W* as follows:

$$W = \sqrt{W_b^2 + W_w^2}.$$
 (27)

We follow a similar approach as in [56,57] to calculate the wind component. The wind-induced component  $W_w$  (N/m) is obtained from the wind pressure  $P_w$  (N/m<sup>2</sup>) and related to the wind velocity  $v_{air}$  as follows:

$$P_w = \frac{1}{2}\rho_{air}v_{air}^2,\tag{28}$$

$$W_w = P_w C_D D \sin^2(\theta_w) \alpha_s, \tag{29}$$

where  $\rho_{air}$  denotes the density of air, *D* denotes the diameter exposed to wind,  $\theta_w$  denotes the angle between the transmission line and wind flow,  $\alpha_s$  denotes the span factor, and  $C_D$  denotes the drag coefficient; the values are shown in Figure 3 for different  $Re_D$ .



**Figure 3.** Values of  $C_D$  for different  $Re_D$ .

## 3.5. Discrete Fourier Transform

The available data for wind and temperature are discrete. We implement the discrete Fourier transform (DFT) and Fourier series to obtain the loading conditions. The following expression defines the DFT:

$$X_k = \sum_{z=0}^{Z-1} a_z e^{-i2\pi \frac{k}{Z}z}$$
(30)

where  $a_z$  denotes the *z*th sample of *a*, and *Z* denotes the total number of the sample. We obtain the mean,  $A_0$ , and frequency coefficients,  $A_z$  and  $B_z$ , for each sample from DFT. The mean and frequency coefficients are used in the Fourier series to obtain the cyclic loading equation using the following:

$$f(t) = A_0 + \sum_{z=1}^{Z/2} [A_z \cos(2\pi z \frac{t}{T}) + B_z \sin(2\pi z \frac{t}{T})]$$
(31)

where *t* is time and *T* represents the time period. The discrete (original data) and continuous data (obtained using the Fourier series analysis) plots of each state are shown in Figures 4–7:



**Figure 4.** Wind and temperature data for Texas. (a) Original wind data. (b) Transformed wind data. (c) Original temperature data. (d) Transformed temperature data.



**Figure 5.** Wind and temperature data for California. (**a**) Original wind data. (**b**) Transformed wind data. (**c**) Original temperature data. (**d**) Transformed temperature data.



**Figure 6.** Wind and temperature data for Michigan. (a) Original wind data. (b) Transformed wind data. (c) Original temperature data. (d) Transformed temperature data.



**Figure 7.** Wind and temperature data for Florida. (**a**) Original wind data. (**b**) Transformed wind data. (**c**) Original temperature data. (**d**) Transformed temperature data.

## 3.6. Reliability of Transmission Lines

In this study, we define a limit state function g(R, S; t) for failure analysis. R denotes the threshold temperature  $\theta_{\text{lim}}$  beyond which failure occurs, S represents the maximum temperature  $\theta_{\text{max}}$  on the transmission line, and t indicates time. We specifically focus on the  $\theta_{\text{max}}$  that the line can withstand before the failure. This makes our function as follows:

$$g(\theta_{\lim}, \theta_{\max}; t) = \theta_{\lim} - \theta_{\max}(t)$$
(32)

While both *R* and *S* are treated as random variables in general cases, *R* is held constant in this study. Only  $\theta_{\text{max}}$  is considered a random variable. Therefore, the probability of failure  $P_f(t)$  is as follows:

$$P_f(t) = P\{g(\theta_{\lim}, \theta_{\max}; t) < 0\}.$$
(33)

## 3.7. Multiphysics Framework

In Figure 8, we present a schematic representation outlining our approach to modeling transmission line failures. The model consists of the main governing equations, including Equations (2), (4), (9), (10), and (20), which describe the interconnection between the thermal properties, electrical properties, and mechanical properties of the system. Additionally, we introduce an environmental module for initial and boundary conditions, as follows:



**Figure 8.** Schematic diagram illustrating the interconnection between four different aspects of the multiphysics framework.

# 4. Deterministic Solution

In this section, we describe the finite element discretization, considering a onedimensional domain, and outline the deterministic solution process. The deterministic solution has two primary objectives: it acts as a black box for stochastic analysis for UQ and provides an interpretable basis for evaluating uncertainty propagation in a higher dimension.

## 4.1. Finite-Element Discretization

We consider the domain length, L = 200 m, and number of elements, N = 1000, for our one-dimensional problem. By multiplying the governing Equations (2), (4), (9), (10), and (20) with a test function w, integrating by parts, and considering the volume differential expressed as the cross-sectional area A(x), we derive their corresponding weak forms as follows:

$$\int_0^L -(1-\varphi_d)^2 Y A(x) \frac{du_d}{dx} \frac{dw}{dx} dx + \int_0^L \gamma g_c A(x) \left(\frac{d\varphi_d}{dx}\right)^2 \frac{dw}{dx} dx + \int_0^L f A(x) w dx = 0, \quad (34)$$

$$\int_{0}^{L} -\gamma g_{c}A(x)\frac{d\varphi_{d}}{dx}\frac{dw}{dx}dx + \int_{0}^{L}A(x)\mathbb{H}wdx - \int_{0}^{L}A\mathbb{H}\varphi_{d}wdx - \int_{0}^{L}\frac{g_{c}A(x)}{\gamma}\varphi_{d}wdx + \int_{0}^{L}\frac{A(x)}{\gamma}\mathcal{F}_{f}wdx = 0,$$
(35)

$$\int_0^L \dot{\mathcal{F}}_f w A(x) dx = \int_0^L \frac{-\rho a (1 - \varphi_d) Y |\frac{du_d}{dx}| (-\varphi_d)}{\gamma} \frac{\theta_c}{\theta_0} w A(x) dx, \tag{36}$$

$$\int_{0}^{L} -\kappa_{m}A(x)\frac{d\theta_{c}}{dx}\frac{dw}{dx}dx + \int_{0}^{L}\sigma_{E}A(x)\left(\frac{dV}{dx}\right)^{2}wdx$$

$$\int_{0}^{L} \log A(x)\left(\frac{dV}{dx}\right)^{2}wdx$$
(37)

$$\int_{0}^{L} h\theta_{c}A_{s}(x)wdx + \int_{0}^{L} h\theta_{a}A_{s}(x)wdx = 0,$$

$$\int_{0}^{L} \sigma_{E}\frac{dV}{dx}\frac{dw}{dx}A(x)wdx = 0,$$
(38)

For each element *k*, we use a linear approximation, as follows:

$$u_d^k = N\hat{u}_d^k,\tag{39}$$

$$\varphi_d^k = N\hat{\varphi}_d^k,\tag{40}$$

$$\mathcal{F}_f{}^k = N \hat{\mathcal{F}}_f{}^k, \tag{41}$$

$$\theta_c^k = N \hat{\theta}_c^k, \tag{42}$$

$$V^k = N\hat{V}^k.$$
(43)

We calculate the finite-element interpolation of spatial derivatives using linear combinations of the derivatives of shape functions:

$$\left(\frac{du_d}{dx}\right)^k = B\dot{u}_d^k,\tag{44}$$

$$\left(\frac{d\varphi_d}{dx}\right)^k = B\hat{\varphi}^k_d,\tag{45}$$

$$\left(\frac{d\theta_c}{dx}\right)^k = B\hat{\theta}_c^k,\tag{46}$$

$$\left(\frac{dV}{dx}\right)^k = B\hat{V}^k,\tag{47}$$

where *N*, *B*,  $\hat{u}_d^k$ ,  $\hat{\varphi}_d^k$ ,  $\hat{\mathcal{F}}_f^{\ k}$ ,  $\hat{\theta}_c^k$ , and  $\hat{V}^k$  are defined as follows:

$$N = \begin{bmatrix} N_1 & N_2 \end{bmatrix},\tag{48}$$

$$B = \begin{bmatrix} N_{1,x} & N_{2,x} \end{bmatrix},\tag{49}$$

$$\hat{u}_d^k = \begin{bmatrix} u_1^k & u_2^k \end{bmatrix},\tag{50}$$

$$\hat{\varphi}_d^k = \begin{bmatrix} \varphi_1^k & \varphi_2^k \end{bmatrix},\tag{51}$$

$$\hat{\mathcal{F}}_{f}^{\ \kappa} = \begin{bmatrix} \mathcal{F}_{1}^{\kappa} & \mathcal{F}_{2}^{k} \end{bmatrix}, \tag{52}$$

$$\hat{\theta}_c^k = \begin{bmatrix} \theta_1^k & \theta_2^k \end{bmatrix},\tag{53}$$

$$\hat{V}^k = \begin{bmatrix} V_1^k & V_2^k \end{bmatrix}. \tag{54}$$

where  $N_1$  and  $N_2$  are linear interpolation functions.

We use a forward Euler method to evolve  $\mathcal{F}_f$ , resulting in the discretization of *k*th element as follows:

$$K_u \hat{u}_d^k = w_u + M \hat{f}^k, \tag{55}$$

$$K_{\varphi}\hat{\varphi}_{d}^{k} = w_{\varphi}, \tag{56}$$

$$M\hat{\mathcal{F}}_{f}^{n+1^{k}} = M\hat{\mathcal{F}}_{f}^{n^{k}} + \Delta t w_{\mathcal{F}},$$
(57)

$$K_{\theta}\hat{\theta}_{c}^{k} = w_{\theta}, \tag{58}$$

$$K_V \hat{V}^k = 0. (59)$$

where the discrete forms are as follows:

$$K_{u} = \int_{k} (1 - N\hat{\varphi}_{d}^{k})^{2} Y A(x) B^{T} B \, dx, \tag{60}$$

$$w_u = \int_k \gamma g_c A(x) (B\hat{\varphi}_d^k)^2 B \, dx,\tag{61}$$

$$M = \int_{k} A(x) N^{T} N \, dx, \tag{62}$$

$$K_{\varphi} = \int_{k} \gamma g_{c} A(x) B^{T} B \, dx + \int_{k} \mathbb{H}A(x) N^{T} N \, dx + \int_{k} \frac{g_{c} A(x)}{\gamma} N^{T} N \, dx, \tag{63}$$

$$w_{\varphi} = \int_{k} \mathbb{H}A(x) N \, dx + \int_{k} \frac{A(x)}{\gamma} N^{T} \hat{\mathcal{F}}_{f}^{n^{k}} N \, dx, \tag{64}$$

$$K_{\theta} = \int_{k} \kappa_{m} A(x) B^{T} B \, dx + \int_{k} h A_{s}(x) N^{T} N \, dx, \tag{65}$$

$$w_{\theta} = \int_{k} \sigma_{E} A(x) \left( B \hat{V}^{k} \right)^{2} N \, dx + \int_{k} h A_{s}(x) \theta_{a} N \, dx, \tag{66}$$

$$K_{v} = \int_{k} (1 - N\hat{\varphi}_{d}^{k})^{2} \sigma_{E,T} A(x) B^{T} B \, dx.$$
(67)

We derive the global forms of these matrices and vectors using standard finite-element assembly procedures.

$$K_u \hat{u}_d = w_u + M \hat{f},\tag{68}$$

$$K_{\varphi}\hat{\varphi}_d = w_{\varphi},\tag{69}$$

$$M\hat{\mathcal{F}}_{f}^{n+1} = M\hat{\mathcal{F}}_{f}^{n} + \Delta t w_{\mathcal{F}},\tag{70}$$

$$K_{\theta}\hat{\theta}_{c} = w_{\theta},\tag{71}$$

$$K_V \hat{V} = 0. \tag{72}$$

The formulation supports a staggered solution scheme at each time step, implemented according to the following Algorithm 1:

Algorithm 1 Multiphysics framework solution.		
Set initial conditions and boundary conditions.		
for Each $\Delta t$ do		
Calculate the horizontal tensile load.		
Solve the governing equation for displacements.		
Update strain energy history.		
Solve the equation for the damage field.		
Update fatigue.		
Solve the heat equation to update the temperature field.		
Solve for the voltage field.		

#### 4.2. Results and Discussion

We consider an all-aluminum conductor (AAC) subjected to cyclic loading conditions under wind, temperature, and current. The details on wind and temperature loading conditions are provided in Section 2. The current loading is parameterized as follows:

$$I(t) = -I_{base} - I_{amp}(\sin 4\pi t), \tag{73}$$

where  $I_{base}$  is the base current, set at 1500 A, and  $I_{amp}$  is the amplitude, set at 100 A, corresponding to the allowable ampacity for a 40 mm aluminum conductor.

We represent the boundary conditions for displacement setting u = 0 and u = H at x = 0 and x = L, respectively. At both ends, we specify  $\frac{d\varphi_d}{dx} = 0$ . Similarly, we apply boundary conditions for the current conservation equation setting V = 0 and  $V = J_d$  at x = 0 and x = L, respectively.

In practice, all materials feature inherent imperfections, so we consider variable crosssection areas in the center of the domain to represent the cumulative effects of these multiple imperfections. We also consider a base scenario where the materials are assumed to have insignificant imperfections, serving as a comparison point with the significant initial damage cases. We define the cross-section area as follows:

$$A(x) = A_0 \left( 1 - \frac{1}{A_\sigma \sqrt{2\pi}} \exp\left(\frac{-(x - L/2)^2}{2A_\sigma^2}\right) \right),$$
(74)

where  $A_0$  denotes the cross-section area without any damage, while  $A_{\sigma}$  indicates the spread-to-depth ratio at the center of the line, representing different levels of damage. Figure 9 illustrates various area profiles based on  $A_{\sigma}$ , as follows:



**Figure 9.** Variable cross-section areas for different values of  $A_{\sigma}$ .

The material parameter values are given in Table 3. The simulations are considered for 60 years with a 6000 run period for which the time-step,  $\Delta t = 0.01$ . Aluminum begins to anneal at temperatures exceeding 366 K [3,58], and rupture occurs when the temperature surpasses 373 K [59]. Consequently, we establish the maximum temperature limit as  $\theta_{\text{lim}} = 373$  K (100 °C). Once this limit is reached, all simulations are stopped.

Table 3. Air, Material, and Geometry parameters.

Parameter	Value	Unit
Domain length L	200	m
Number of elements N	1000	
Initial Tension $H_0$	40	kN
Diameter D	0.04	m
Young modulus Y	69	GPa
Damage layer width $\gamma$	0.02	m
Fracture energy $g_c$	10	kN/m
Density $\rho_m$	2700	kg/m <sup>3</sup>
Aging coefficient <i>a</i>	$1 imes 10^{-10}$	m <sup>5</sup> /(y kg)
Thermal conductivity $\kappa_m$	237	W/(m K)
Electrical conductivity $\sigma_{E,0}$	$3.77  imes 10^7$	S/m
Temperature coefficient $\alpha_l$	$3.9 imes10^{-3}$	$\mathrm{K}^{-1}$
Density of air $\rho_{air}$	1.225	kg/m <sup>3</sup>
Kinematic viscosity of air $v_{air}$	$15  imes 10^{-6}$	$m^2/s$
Thermal conductivity of air $\kappa_{air}$	0.0295	W/(m K)
Prandtl Pr	0.71	

We begin our analysis by examining the evolution of field quantities in the Texas scenario, using the parameters in Table 3. Every five years, we plot the evolution of field variables along the length of the transmission line as shown in Figure 10. We note that damage typically begins and builds up in areas with smaller cross-sections, causing increased temperatures and notable disturbances in the voltage fields. As damage and temperature rise, the voltage drop along the line also increases over time due to higher electrical resistance in the line.

Next, we study the effect of  $A_{\sigma}$  on the maximum values of field variables. Figure 11 shows that the presence of damage affects the evolution of field variables at the center of the line.



**Figure 10.** Evolution of field variables. (**a**) Damage evolution along the line. (**b**) Fatigue evolution along the line. (**c**) Temperature evolution along the line. (**d**) Voltage drops along the line.



**Figure 11.** Effect of initial damage on maximum field values over time. (a) Maximum damage. (b) Maximum fatigue. (c) Maximum Temperature. (d) Maximum voltage drop.

Moreover, we compare the life span of transmission lines for each specific scenario varying the cross-sectional area of damage  $A_{\sigma}$  under respective loading conditions in

Figure 12. In the Texas scenario, we observe the life span of the transmission line reduces from 51 years to 44 years with moderate damage. However, with severe damage, the life span reduces to 28 years, indicating how detrimental initial damage could be to the longevity of the material. Similarly, in the California scenario, the lifespan of the transmission line reduces from 52 years to 26 years under severe damage. This reduction is due to the state's high temperatures and low wind speeds, significantly lowering the convective cooling effect. Consequently, in the absence of the cooling effect, the accumulation of damage raises the material temperature, leading to overheating and early failure. In the case of the Michigan scenario, the state shows the most fluctuating temperature and wind over a year, which progresses the damage, raising the temperature of the material and leading to early failure under insignificant damage conditions among the four states. However, due to the high wind speed and low temperature enhancing convective cooling during some seasons, the life span under severe damage is higher than the California scenario. The Florida scenario shows high temperatures over a year, similar to Texas and California. However, the high wind speed in the state enhances convective cooling, reducing the early failure of the transmission line. Overall, every scenario shows a significant reduction in the life span of the transmission line under severe damage.



Figure 12. The failure of transmission lines for different values of initial damage. (a) Texas. (b) California. (c) Michigan. (d) Florida.

## 5. Stochastic Solution

In this section, we address the challenge of performing UQ, SA, and the probability of failure ( $P_f$ ) on the model treated as a black box. We focus on the conductor temperature as our quantity of interest (QoI). The non-intrusive nature of our approach is particularly beneficial, allowing us to apply the same methods of deterministic solutions to stochastic problems without altering the model's governing equations.

We employ the PCM for three main purposes. First, we use it to compute the QoI. Second, we conduct global sensitivity analysis by calculating Sobol sensitivity indices ( $S_i$ ). These indices highlight the relative importance of each model parameter by quantifying the contribution to the variance of the QoI. Lastly, we use PCM to facilitate the computation of the probability of failure in a straightforward fashion by calculating the expectation of the Bernoulli random variable.

## 5.1. Uncertainty Quantification

We followed the approach outlined in [27] to conduct UQ using PCM. We assume  $(\Omega_s, \mathcal{G}, \mathbb{P})$  as a complete probability space, where  $\Omega_s$  denotes the space of outcomes  $\omega$ ,  $\mathcal{G}$  is the  $\sigma$ -algebra, and  $\mathbb{P}$  is a probability measure mapping  $\mathcal{G}$  to the interval [0,1]. In our model, we treat the material and loading parameters as random variables with two different sets,  $\xi_m(\omega)$  and  $\xi_l(\omega)$ . We model our QoI temperature as random variables and denoted random parameters as  $\xi = \xi(\omega)$ .

Our QoI, denoted as  $Q_{oI}$ , involves calculating the expectation  $\mathbb{E}[Q_{oI}(x, t; \xi)]$  as

$$\mathbb{E}[Q_{oI}(x,t;\xi)] = \int_{a}^{b} Q_{oI}(x,t;\xi)\rho(\xi)d\xi,$$
(75)

where the probability density function (PDF) of  $\xi$ , denoted as  $\rho(\xi)$ , is evaluated by mapping the physical parametric space to the standard space [-1, 1], followed by integration using the Gauss quadrature. This transformation allows the integral to be expressed within this standard interval as follows:

$$\mathbb{E}[Q_{oI}(x,t;\xi)] = \int_{-1}^{1} Q_{oI}(x,t;\xi(\eta))\rho(\xi(\eta))Jd\xi(\eta),$$
(76)

where  $J = d\xi/d\eta$  represents the Jacobian of the transformation. To approximate the expectation, we use polynomial interpolation in the stochastic space, expressed as  $\hat{Q}_{oI}(x,t;\xi)$ , which approximates the exact solution as follows:

$$\mathbb{E}[Q_{oI}(x,t;\xi)] \approx \int_{-1}^{1} \hat{Q}_{oI}(x,t;\xi(\eta))\rho(\xi(\eta)) Jd\xi(\eta).$$
(77)

We use Lagrange polynomials  $L_i(\xi)$  to interpolate the solution in the stochastic space, as follows:

$$\hat{Q}_{oI}(x,t;\xi) = \sum_{i=1}^{l} Q_{oI}(x,t;\xi_i) L_i(\xi),$$
(78)

which satisfies the Kronecker delta property at the interpolation points as follows:

$$L_i(\xi_j) = \delta_{ij}.\tag{79}$$

We substitute the polynomial approximation from Equation (78) into Equation (77) and use the quadrature rule to approximate the integral and compute the expectation as follows:

$$\mathbb{E}[Q_{oI}(x,t;\xi)] \approx \sum_{m=1}^{n} w_m \rho(\xi(\eta)) J \sum_{i=1}^{l} Q_{oI}(x,t;\xi(\eta)) L_i(\xi(\eta)),$$
(80)

where  $\eta_m$  and  $w_m$  are coordinates and weights, respectively, for each integration point m = 1, 2, ..., n. We choose the same collocation and integration points using Equation (79) and simplify Equation (80) as follows:

$$\mathbb{E}[Q_{oI}(x,t;\xi)] = \sum_{m=1}^{n} w_m \rho(\xi_m(\eta_m)) J Q_{oI}(x,t;\xi_m(\eta_m)).$$
(81)

We use  $\xi_m(\eta_m) = a + \frac{(b-a)}{2}(\eta_m + 1)$  as linear affine mapping that calculates the Jacobian as J = (b-a)/2. Finally, we approximate the integration and rewrite as a

summation over the collocation points, assuming a uniform distribution for the parameters throughout the interval [a, b], with  $\rho(\xi) = 1/(b-a)$ . The expectation is as follows:

$$\mathbb{E}[Q_{oI}(x,t;\xi)] = \frac{1}{2} \sum_{m=1}^{n} w_m Q_{oI}(x,t;\xi_m).$$
(82)

Similar to the MC method, the standard deviation is computed as follows:

$$\sigma[Q_{oI}(x,t;\xi)] = \sqrt{\frac{1}{2} \sum_{m=1}^{n} w_m (Q_{oI}(x,t;\xi_m) - \mathbb{E}[Q_{oI}(x,t;\xi)])^2}.$$
(83)

For the higher dimensions, Equation (75) includes additional integration that reduces to the following:

$$\mathbb{E}\Big[Q_{oI}(x,t;\xi^1,\ldots,\xi^k)\Big] = \mathbb{E}_{PCM}\Big[Q_{oI}(x,t;\xi^1,\ldots,\xi^k)\Big]$$
$$\approx \sum_{m=1}^n \cdots \sum_{l=1}^L w_m \ldots w_l \rho(\xi_m) \ldots \rho(\xi_l) J_m \ldots J_l Q_{oI}(x,t;\xi_m^1,\ldots,\xi_l^k) \tag{84}$$

In the expanded version of PCM for higher dimensions, we have *k* summations corresponding to each dimension in the random space. For simplicity, we represent the expectation as  $\mathbb{E}\left[Q_{oI}(x,t;\xi^1,\ldots,\xi^k)\right] = \mathbb{E}[Q_{oI}]$ . We then formulate the standard deviation as follows:

$$\sigma \Big[ Q_{oI}(x,t;\xi^1,\ldots,\xi^k) \Big] = \sigma_{PCM} \Big[ Q_{oI}(x,t;\xi^1,\ldots,\xi^k) \Big]$$
$$\approx \sqrt{\sum_{m=1}^n \cdots \sum_{l=1}^L w_m \ldots w_l \rho(\xi_m) \ldots \rho(\xi_l) J_m \ldots J_l \left( Q_{oI}(x,t;\xi^1_m,\ldots,\xi^k_l) - \mathbb{E}[Q_{oI}] \right)^2}.$$
(85)

Here, we assume discretization in the parametric space is isotropic, and the random variables are mutually independent. Additionally, PCM is sufficient for the model in this study, as it involves fewer than six dimensions. However, this approach becomes inefficient for models with more than six dimensions due to the exponential increase in the number of simulations required by the tensor product. The problem of dimensionality can be reduced using Smolyak sparse grids [46], Principal Component Analysis [60], active subspaces methods [47], and low-rank approximations [61].

## 5.2. Sensitivity Analysis

We examine the sensitivity analysis using Sobol indices [62]. For details on the derivation, we refer to the work by Saltelli et al. [63]. In our global sensitivity analysis, each parameter is denoted as  $\xi^u$ , where u = 1, 2, ..., v. We evaluate the influence of  $\xi^u$  on the variance of the QoI as follows:

$$V_{\xi^{u}}\left(\mathbb{E}_{\tilde{\xi}^{u}}(Q_{oI}|\xi^{u})\right)$$
(86)

where  $\tilde{\xi}^{u}$  represents all possible values of the random parameters except for  $\xi^{u}$ , which is held fixed. Equation (86) describes the process of computing the expected value of  $Q_{oI}$  with  $\xi^{u}$  fixed. Based on the law of total variance, we have the following:

$$V_{\tilde{\xi}^{u}}\left(\mathbb{E}_{\tilde{\xi}^{u}}(Q_{oI}|\xi^{u})\right) + \mathbb{E}_{\tilde{\xi}^{u}}\left(V_{\tilde{\xi}^{u}}(Q_{oI}|\xi^{u})\right) = V(Q_{oI})$$

$$\tag{87}$$

We normalize Equation (87) to obtain the first-order  $S_i$ . This index quantifies the impact of the random variable  $\xi^u$  on the total variance and is calculated as follows:

$$S_{i} = \frac{V_{\bar{\xi}^{u}} \left( \mathbb{E}_{\bar{\xi}^{u}} (U | \xi^{u}) \right)}{V(U)}$$
(88)

 $S_i$  only quantifies the first-order effects of the variable  $\xi^u$  on the variance. Following normalization, the sum of all  $S_i$  values is less than 1, with the remaining portion representing interactions among the parameters, which are not considered in this paper.

#### 5.3. Probability of Failure

In this section, we calculate the  $P_f$  over time. Traditionally, MC methods are employed in reliability analysis to compute the probability of failure [64]. Stochastic collocation methods are used for similar analysis; however, the method primarily yields the moments of a limit state function g(R, S). To compute  $P_f$ , these moments must be transformed into a PDF, which can be achieved using various techniques such as the method of moments [65,66], polynomial chaos [67,68], Gaussian transformations [69], or entropy optimization methods [70]. However, these processes add a layer of complexity.

In this study, we present an alternative way that utilizes the efficiency of the PCM to compute the probability of failure ( $P_f$ ). Rather than calculating and approximating PDF to determine P(g < 0), we simplify the process by transforming g into a Bernoulli random variable  $h_B$  with coefficient  $p_h$ .

The transformation of *g* to  $h_B$  is defined as follows:

$$h_B = \begin{cases} 0, & \text{if } g \ge 0, \\ 1, & \text{otherwise.} \end{cases}$$
(89)

In single realization, we obtain  $h_B$ , which remains zero until the point where  $\theta_{max}$  exceeds the threshold  $\theta_{lim}$ , at which  $h_B$  becomes one and remains at this value. This represents a step function for each realization.

At a fixed time-step, considering the expectation of  $h_B$ , due to the smoothness of the QoI,  $h_B$  can be assumed as a real value between 0 and 1 which reflects the probability of the maximum temperature exceeding the limit up to that time.

#### 5.4. Results and Discussion

In the deterministic case, we understood how varying initial damage under cyclic loading conditions influences the lifespan of the transmission line. Now, we shift our focus to understanding the impact of uncertainty in the parameters on maximum temperature. For this analysis, we model the parametric uncertainty with a uniform distribution, where each parameter varies by 10% around its mean value.

#### 5.4.1. Texas Scenario

We first consider the Texas scenario as our baseline for conducting preliminary analyses on the uncertainties associated with parametric material properties and input loading conditions. Our primary goal is to analyze UQ and perform SA on factors influencing the maximum temperature. We first focus on material parameters, represented by the set  $\xi_m(\omega) = \{A_\sigma(\omega), \gamma(\omega), g_c(\omega), a(\omega)\}$ . These parameters are selected due to their inherent measurement inaccuracies or their assumptions within our modeling. We assume the rest of the material parameters to be deterministic. Following the analysis of material parameters, we then examine the uncertainty in loading conditions, denoted by the set  $\xi_l(\omega) = \theta_{base}(\omega), w_{base}(\omega), I_{base}(\omega), I_{amp}(\omega)$ . From both analyses using 5 PCM points per dimension, we identify two material parameters and three loading parameters that are most influential according to global SA results.

We begin by examining the uncertainty in the material parameters from the set  $\xi_m(\omega)$ . We analyze the expectation and standard deviation of the temperature field over time, as presented in Figure 13. Similarly to the deterministic case, the location at the conductor with initial damage has the maximum temperature and standard deviation. Considering this, we plot the evolution of the expected maximum temperature and the standard deviation at the center of the transmission line, where both the maximum temperature and standard deviation increased over time, as shown in Figure 14. We plot the evolution of the expected maximum temperature and the standard deviation similarly for loading parametric set  $\xi_l(\omega)$  in Figure 15.



**Figure 13.** Expected temperature and standard deviation of temperature under the material parametric space  $\xi_m(\omega)$  in the Texas scenario. (a) Expected temperature. (b) Temperature standard deviation.



**Figure 14.** Expected maximum temperature and standard deviation of maximum temperature under material parametric space  $\xi_m(\omega)$  in the Texas scenario. (a) Maximum expected temperature. (b) Standard deviation of maximum temperature.



**Figure 15.** Expected maximum temperature and standard deviation of maximum temperature under the parametric space  $\xi_l(\omega)$  in the Texas scenario. (a) Maximum expected temperature. (b) Standard deviation of maximum temperature.

We use Equation (88) to calculate the Sobol indices  $S_i$  and shown in Figure 16 for material parameters set  $\xi_m(\omega)$  and loading parameters set  $\xi_l(\omega)$ . Initially,  $A_\sigma$  has a significant impact, but it is later surpassed by  $g_c$  and a. The parameter  $I_{base}$ , among the loading parameters, sets  $\xi_l(\omega)$  from the initial state and emerges as the most crucial factor influencing the uncertainty of  $\theta_{max}$  due to its direct impact on Joule heating. Meanwhile, the wind base parameter,  $w_{base}$ , and the temperature base parameter also play significant roles due to the interplay between Joule heating and convective cooling:



**Figure 16.** Sensitivity index  $S_i$  for the Texas scenario for material parameters,  $\xi_m(\omega)$ , and loading parameters,  $\xi_l(\omega)$ . (a) Material parameters. (b) External loading.

We combine the two most influential material parameters with the three most influential loading parameters to form a new set,  $\xi_1(\omega) = g_c(\omega), a(\omega), \theta_{base}(\omega), w_{base}(\omega), I_{base}(\omega)$ , and conduct a sensitivity analysis for Texas. Figure 17 shows that initially, the loading conditions have a more substantial impact than material parameters, but over time, as aging effects become more significant, the relative importance of material parameters increases.



**Figure 17.** Expected maximum temperature, standard deviation of maximum temperature, and sensitivity index over time under parametric space  $\xi(\omega)$  in the Texas scenario. (a) Maximum expected temperature. (b) Standard deviation of maximum temperature. (c) Sensitivity index.

## 5.4.2. California, Michigan, and Florida

In California, Michigan, and Florida scenarios, we combine the two most influential material parameters and the three most influential loading parameters to obtain a new parametric set,  $\xi(\omega) = \{g_c(\omega), a(\omega), \theta_{base}(\omega), w_{base}(\omega), I_{base}(\omega)\}$ , to perform the SA.

In the California scenario, when the time series for maximum temperature is truncated at the earliest occurrence of failure, the analysis reveals that failures happen within a span shorter than five years. However, the pattern in Figure 18 aligns closely with observations from the Texas Scenario. The consistency in the pattern highlights the dominant influence of the current base parameter,  $I_{base}$ , which is intensified by low wind speeds that reduce convective cooling. As a result, the conductor temperature rapidly increases to critical thresholds, driven by the Joules heating in the absence of effective cooling:



**Figure 18.** Expected maximum temperature, standard deviation of maximum temperature, and sensitivity index over time under parametric space  $\xi(\omega)$  in the California scenario. (a) Maximum expected temperature. (b) Standard deviation of maximum temperature. (c) Sensitivity index.

In the Michigan scenario, shown in Figure 19,  $I_{base}$  initially has the most significant impact on the failure of the transmission line. Over time, however,  $g_c$  becomes increasingly influential, eventually surpassing the impact of  $I_{base}$ . Historical data from Michigan shows that the state experiences higher wind speeds compared to the other states analyzed. Although the wind has a convective effect, the impact of increased mechanical load on the cable is more significant, accelerating the transmission line's aging:



**Figure 19.** Expected maximum temperature, standard deviation of maximum temperature, and sensitivity index over time under parametric space  $\xi(\omega)$  in the Michigan scenario. (a) Maximum expected temperature. (b) Standard deviation of maximum temperature. (c) Sensitivity index.

In the Florida scenario, as illustrated in Figure 20,  $I_{base}$  initially shows the most significant influence on the transmission line's failure. However, over time, the fracture energy parameter,  $g_c$ , becomes more dominant. Historical data considered in this study show that Florida experiences high temperatures throughout the year. Despite higher wind speeds, the effectiveness of convective cooling is diminished due to the high ambient temperatures.



**Figure 20.** Expected maximum temperature, standard deviation of maximum temperature, and sensitivity index over time under parametric space  $\xi(\omega)$  in the Florida scenario. (a) Maximum expected temperature. (b) Standard deviation of maximum temperature. (c) Sensitivity index.

In this section, we first calculate the expected value of *h*, with the application of PCM considering n = 5. The results are displayed in Figure 21 for the reference mean parameter values from the parameter sets  $\xi_1(\omega)$ ,  $\xi_2(\omega)$ ,  $\xi_3(\omega)$ , and  $\xi_4(\omega)$ . Under the consideration of four specific states, the probability of failure curve for California initially shows the early higher possibility of occurrence of failure. In contrast, the Florida scenario shows the failure curve on the right shift, indicating a lower possibility of failure among the four states.



Figure 21. Probability of failure for Texas, California, Michigan, and Florida.

Finally, we analyze the impact of varying mean values of five influential parameters  $\xi_1(\omega) = g_c(\omega), a(\omega), \theta_{base}(\omega), w_{base}(\omega), I_{base}(\omega)$  under three different levels of  $A_\sigma$  in the timeseries probability of failure ( $P_f$ ) for each scenario. The comparison includes a baseline scenario representing a transmission line with minimal initial damage. Unlike previous sensitivity analyses of parameters, which provided insights into levels of importance relative to their uncertainty, this section focuses on the effects of changing the baseline values of influential parameters on the  $P_f$  curves under minimal, moderate, and severe initial damage.

In each scenario, we observe three distinct curve shapes representing different levels of damage, as depicted in Figure 22. The analysis shows that in the presence of moderate initial damage, the chance of failure increases from 20% to 60% after 45 years in the Texas, California, and Michigan scenarios. In Florida, the probability of failure increases from 20% to 60% after 45 years. In severe initial damage, the probability of failure reaches 100% well before the failure initiates in a minimal damage case. All the states show the life span of the transmission lines to be around 35 to 40 years under severe damage. This indicates how detrimental the initial damage could be to the life span of the transmission line. Among all the four scenarios, the chance of failure was initiated early in the case of California, which has low wind and high ambient temperature, reducing the impact of convective cooling. Also, the plots reveal that the probability of failure is around 90% in 60 years, indicating a 10% chance the transmission line will last more than 60 years. However, in the case of Michigan, as it has higher fluctuations in wind and temperature, the chance of failure is maximum at 60 years compared to the three states.



Figure 22. Cont.



**Figure 22.** The probability of failure of transmission lines for different values of initial damage (shown in legends). (a) Texas. (b) California. (c) Michigan. (d) Florida.

# 6. Verification and Analysis of the Model

As the model involves the coupling of several governing equations, obtaining an analytical solution is not feasible. Therefore, we perform convergence analysis using a refined PCM solution as our reference:

$$\epsilon = \frac{\|\theta - \theta_{ref}\|_2}{\|\theta_{ref}\|_2}.$$
(90)

In our convergence study, we defined the problem as a 1D problem, considering the most influential parameter  $I_{base}$  as the only model parameter. We considered 100 collocation points as the reference solution and calculated the relative errors until the earliest failure, considering the Texas Scenario as a reference. The comparison is graphically represented in Figure 23 to highlight the accuracy of PCM relative to Monte Carlo methods:



Figure 23. Error estimation using PCM and MC method plots (a) PCM; (b) MC.

The probabilistic collocation method achieved significantly lower errors, approximately two orders of magnitude less, compared to 10,000 Monte Carlo simulations using just 5 collocation points.

## 7. Conclusions

We developed an integrated model to study the reliability of transmission lines across four specific states in the US: Texas, California, Michigan, and Florida. We considered the historical wind and temperature data of each state. The model integrates the fatigue and damage phase-field model with a thermal model that accounts for Joule heating and convective cooling, as well as an electrical model that addresses how accumulated damage and temperature-induced resistance lead to voltage drops along the transmission line.

We used the finite element method to solve all the governing equations. We studied four different states of the US, Texas, California, Michigan, and Florida, to understand the long-term behavior of the overhead power lines in the presence of minimal, moderate, and severe damage. We used discrete Fourier analysis to obtain the continuous loading condition from the discrete historical data of wind and temperature. To reduce the complexity, we parameterized the current loading. Subsequently, we implemented the PCM for UQ, SA, and probability of failure assessments.

The deterministic solution revealed how the temperature in the conductor under cyclic wind, air temperature, and current loading in the presence of initial damage affects the overall longevity of the transmission lines. Even the moderate initial damage significantly reduced the material's life expectancy. In the global sensitivity analysis using PCM, we identified that  $I_{base}$  initially had the most significant effect, which was later surpassed by  $g_c$  over time. Additionally, exploring the further application of PCM, we studied the probability of failure for each specific scenario, varying the most influential parameters for three damage cases. The analysis revealed that the chance of failure was significantly increased even with moderate initial damage.

Overall, this research provides critical insights into how initial damage significantly reduces the lifespan of transmission lines—an aspect often overlooked in previous studies. By highlighting key parameters that most influence transmission line reliability, this predictive model offers direct applications for improving transmission line design and enhancing resilience. Furthermore, by incorporating realistic environmental data, the model can be adapted to various regions, making it a flexible and practical tool for assessing transmission line reliability across diverse climates and environmental conditions.

However, the model incorporates simplifying assumptions such as static conditions and the one-dimensional domain of the cable, which could be further explored in the future. Even the integration of real current data for specific states, along with the effects of solar radiation and solar heat gain, could enhance the predictive accuracy of failure probabilities. All these modifications shift the dependency from abstract stochastic process modeling to a more robust physics-based approach, significantly improving the reliability assessments of the transmission line.

Author Contributions: Conceptualization, P.K. and M.Z.; methodology, P.K.; software, P.K.; validation, P.K., M.Z. and M.N.; formal analysis, P.K., M.Z. and M.N.; investigation, P.K., M.Z. and M.N.; resources, M.Z.; data curation, P.K.; writing—original draft preparation, P.K.; writing—review and editing, M.Z. and M.N.; visualization, P.K.; supervision, M.Z. and M.N.; project administration, M.Z.; funding acquisition, M.Z. All authors have read and agreed to the published version of the manuscript.

**Funding:** This research was funded by an ARO Young Investigator Program (YIP) award (W911NF-19-1-0444) and an NSF award (DMS-1923201).

Data Availability Statement: Data can be provided upon request.

**Acknowledgments:** All authors would like to thank the Institute for Cyber-Enabled Research (ICER) at Michigan State University for providing HPC resources and services.

**Conflicts of Interest:** The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

#### Abbreviations

The following abbreviations are used in this manuscript:

- PCM probability collocation method
- UQ uncertainty quantification
- SA sensitivity analysis
- QoI quantity of interest

MC	Monte Carlo
PDF	probability density function
PDE	partial differential equation
ODE	ordinary differential equation
AAC	all-aluminum conductor
ACSR	aluminum conductor steel-reinforced
DFT	discrete Fourier transform
DDD	discrete dislocation dynamics
ΤX	Texas
CL	California
MI	Michigan
FL	Florida

# References

- 1. DOE. *Economic Benefits of Increasing Electric Grid Resilience to Weather Outages*; Executive Office of the President: Washington, DC, USA , 2013.
- Eaton. Blackout Tracker Power Outage Annual Report 2018. Available online: https://www.eaton.com/content/dam/eaton/ products/backup-power-ups-surge-it-power-distribution/backup-power-ups/blackout-tracker-/blackout-tracker-annualreport-2018-mz153043en.pdf (accessed on 13 November 2024).
- Hathout, I.; Callery, K.; Trac, J.; Hathout, T. Impact of thermal stresses on the end of life of overhead transmission conductors. In Proceedings of the 2018 IEEE Power & Energy Society General Meeting (PESGM), Portland, OR, USA, 5–10 August 2018; pp. 1–5.
- 4. Alminhana, F.; Mason, M.; Albermani, F. Transmission line failure propagation under extreme wind scenarios: extratropical cyclones. *Eng. Struct.* **2023**, *284*, 115942. [CrossRef]
- 5. Dua, A.; Clobes, M.; Höbbel, T. Dynamic analysis of overhead transmission line under turbulent wind loading. *Electron. J. Struct. Eng.* **2015**, *15*, 46–54. [CrossRef]
- Stengel, D.; Mehdianpour, M. Finite element modelling of electrical overhead line cables under turbulent wind load. *J. Struct.* 2014, 2014, 421587. [CrossRef]
- Hamada, A.; El Damatty, A. Behaviour of guyed transmission line structures under tornado wind loading. *Comput. Struct.* 2011, 89, 986–1003. [CrossRef]
- 8. Guo, Y.; Chen, R.; Shi, J.; Wan, J.; Yi, H.; Zhong, J. Determination of the power transmission line ageing failure probability due to the impact of forest fire. *IET Gener. Transm. Distrib.* **2018**, *12*, 3812–3819. [CrossRef]
- 9. Rossi, A.; Jubayer, C.; Koss, H.; Arriaga, D.; Hangan, H. Combined effects of wind and atmospheric icing on overhead transmission lines. *J. Wind Eng. Ind. Aerodyn.* 2020, 204, 104271. [CrossRef]
- Bendík, J.; Cenkỳ, M.; Eleschova, Ž.; Beláň, A.; Cintula, B. Influence of different weather conditions on the maximum load current of overhead power lines. In Proceedings of the 2018 19th International Scientific Conference on Electric Power Engineering (EPE), Brno, Czech Republic, 16–18 May 2018; pp. 1–4.
- Pytlak, P.; Musilek, P.; Lozowski, E. Precipitation-based conductor cooling model for dynamic thermal rating systems. In Proceedings of the 2009 IEEE Electrical Power & Energy Conference (EPEC), Montreal, QC, Canada, 22–23 October 2009; pp. 1–7.
- Castro, P.; Arroyo, A.; Martinez, R.; Manana, M.; Domingo, R.; Laso, A.; Lecuna, R. Study of different mathematical approaches in determining the dynamic rating of overhead power lines and a comparison with real time monitoring data. *Appl. Therm. Eng.* 2017, 111, 95–102. [CrossRef]
- 13. Zhu, M.X.; Song, H.G.; Li, J.C.; Yu, Q.C.; Chen, J.M. Phase-field modeling of electric-thermal breakdown in polymers under alternating voltage. *IEEE Trans. Dielectr. Electr. Insul.* 2020, 27, 1128–1135. [CrossRef]
- 14. Woldman, A.Y.; Landis, C.M. Thermo-electro-mechanical phase-field modeling of paraelectric to ferroelectric transitions. *Int. J. Solids Struct.* **2019**, *178*, 19–35. [CrossRef]
- 15. Shen, Z.H.; Wang, J.J.; Jiang, J.Y.; Huang, S.X.; Lin, Y.H.; Nan, C.W.; Chen, L.Q.; Shen, Y. Phase-field modeling and machine learning of electric-thermal-mechanical breakdown of polymer-based dielectrics. *Nat. Commun.* **2019**, *10*, 1843. [CrossRef]
- 16. Tanné, E.; Li, T.; Bourdin, B.; Marigo, J.J.; Maurini, C. Crack nucleation in variational phase-field models of brittle fracture. *J. Mech. Phys. Solids* **2018**, *110*, 80–99. [CrossRef]
- 17. Li, B.; Peco, C.; Millán, D.; Arias, I.; Arroyo, M. Phase-field modeling and simulation of fracture in brittle materials with strongly anisotropic surface energy. *Int. J. Numer. Methods Eng.* **2015**, *102*, 711–727. [CrossRef]
- 18. Borden, M.J.; Hughes, T.J.; Landis, C.M.; Verhoosel, C.V. A higher-order phase-field model for brittle fracture: Formulation and analysis within the isogeometric analysis framework. *Comput. Methods Appl. Mech. Eng.* **2014**, 273, 100–118. [CrossRef]
- 19. Miehe, C.; Hofacker, M.; Welschinger, F. A phase field model for rate-independent crack propagation: Robust algorithmic implementation based on operator splits. *Comput. Methods Appl. Mech. Eng.* **2010**, *199*, 2765–2778. [CrossRef]
- Miehe, C.; Welschinger, F.; Hofacker, M. Thermodynamically consistent phase-field models of fracture: Variational principles and multi-field FE implementations. *Int. J. Numer. Methods Eng.* 2010, *83*, 1273–1311. [CrossRef]
- 21. de Moraes, E.A.B.; Salehi, H.; Zayernouri, M. Data-driven failure prediction in brittle materials: A phase field-based machine learning framework. *J. Mach. Learn. Model. Comput.* **2021**, *2*, 1–20. [CrossRef]

- 22. Seiler, M.; Linse, T.; Hantschke, P.; Kästner, M. An efficient phase-field model for fatigue fracture in ductile materials. *Eng. Fract. Mech.* **2020**, 224, 106807. [CrossRef]
- 23. Miehe, C.; Aldakheel, F.; Raina, A. Phase field modeling of ductile fracture at finite strains: A variational gradient-extended plasticity-damage theory. *Int. J. Plast.* **2016**, *84*, 1–32. [CrossRef]
- 24. Kuhn, C.; Noll, T.; Müller, R. On phase field modeling of ductile fracture. GAMM-Mitteilungen 2016, 39, 35–54. [CrossRef]
- 25. Ambati, M.; Kruse, R.; De Lorenzis, L. A phase-field model for ductile fracture at finite strains and its experimental verification. *Comput. Mech.* **2016**, *57*, 149–167. [CrossRef]
- Ambati, M.; Gerasimov, T.; De Lorenzis, L. Phase-field modeling of ductile fracture. *Comput. Mech.* 2015, 55, 1017–1040. [CrossRef]
- Barros de Moraes, E.A.; Zayernouri, M.; Meerschaert, M.M. An integrated sensitivity-uncertainty quantification framework for stochastic phase-field modeling of material damage. *Int. J. Numer. Methods Eng.* 2021, 122, 1352–1377. [CrossRef]
- 28. Boldrini, J.; de Moraes, E.B.; Chiarelli, L.; Fumes, F.; Bittencourt, M. A non-isothermal thermodynamically consistent phase field framework for structural damage and fatigue. *Comput. Methods Appl. Mech. Eng.* **2016**, *312*, 395–427. [CrossRef]
- de Moraes, E.A.B.; Suzuki, J.L.; Zayernouri, M. Atomistic-to-meso multi-scale data-driven graph surrogate modeling of dislocation glide. Comput. Mater. Sci. 2021, 197, 110569. [CrossRef]
- Chhetri, S.; de Moraes, E.; Naghibolhosseini, M.; Zayernouri, M. A Comparative Study of Dislocation Dynamics in Ductile and Brittle Crystalline Materials. In Proceedings of the 2023 International Conference on Modeling, Simulation & Intelligent Computing (MoSICom), Dubai, United Arab Emirates, 7–9 December 2023; pp. 438–441.
- de Moraes, E.A.B.; D'Elia, M.; Zayernouri, M. Machine learning of nonlocal micro-structural defect evolutions in crystalline materials. *Comput. Methods Appl. Mech. Eng.* 2023, 403, 115743. [CrossRef]
- 32. Suzuki, J.L.; Naghibolhosseini, M.; Zayernouri, M. A General Return-Mapping Framework for Fractional Visco-Elasto-Plasticity. *Fractal Fract.* **2022**, *6*, 715. [CrossRef]
- 33. Suzuki, J.; Zayernouri, M.; Bittencourt, M.; Karniadakis, G. Fractional-order uniaxial visco-elasto-plastic models for structural analysis. *Comput. Methods Appl. Mech. Eng.* 2016, 308, 443–467. [CrossRef]
- 34. Suzuki, J.; Zhou, Y.; D'Elia, M.; Zayernouri, M. A thermodynamically consistent fractional visco-elasto-plastic model with memory-dependent damage for anomalous materials. *Comput. Methods Appl. Mech. Eng.* **2021**, 373, 113494. [CrossRef]
- Suzuki, J.L.; Gulian, M.; Zayernouri, M.; D'Elia, M. Fractional modeling in action: A survey of nonlocal models for subsurface transport, turbulent flows, and anomalous materials. J. Peridynamics Nonlocal Model. 2023, 5, 392–459. [CrossRef]
- Kharazmi, E.; Zayernouri, M. Operator-based uncertainty quantification of stochastic fractional partial differential equations. J. Verif. Valid. Uncertain. Quantif. 2019, 4, 041006. [CrossRef]
- Fishman, G. Monte Carlo: Concepts, Algorithms, and Applications; Springer Science & Business Media: Berlin/Heidelberg, Germany, 2013.
- 38. Smith, R.C. Uncertainty Quantification: Theory, Implementation, and Applications; SIAM: Philadelphia, PA, USA, 2013.
- Xiu, D.; Karniadakis, G.E. Modeling uncertainty in steady state diffusion problems via generalized polynomial chaos. *Comput. Methods Appl. Mech. Eng.* 2002, 191, 4927–4948. [CrossRef]
- 40. Xiu, D.; Karniadakis, G.E. The Wiener–Askey polynomial chaos for stochastic differential equations. *SIAM J. Sci. Comput.* **2002**, 24, 619–644. [CrossRef]
- 41. Stefanou, G. The stochastic finite element method: Past, present and future. *Comput. Methods Appl. Mech. Eng.* 2009, 198, 1031–1051. [CrossRef]
- 42. Babuška, I.; Tempone, R.; Zouraris, G.E. Solving elliptic boundary value problems with uncertain coefficients by the finite element method: The stochastic formulation. *Comput. Methods Appl. Mech. Eng.* 2005, 194, 1251–1294. [CrossRef]
- Babuska, I.; Tempone, R.; Zouraris, G.E. Galerkin finite element approximations of stochastic elliptic partial differential equations. SIAM J. Numer. Anal. 2004, 42, 800–825. [CrossRef]
- 44. Babuška, I.; Nobile, F.; Tempone, R. A stochastic collocation method for elliptic partial differential equations with random input data. *SIAM J. Numer. Anal.* 2007, 45, 1005–1034. [CrossRef]
- 45. Xiu, D.; Hesthaven, J.S. High-order collocation methods for differential equations with random inputs. *SIAM J. Sci. Comput.* 2005, 27, 1118–1139. [CrossRef]
- 46. Smolyak, S.A. Quadrature and interpolation formulas for tensor products of certain classes of functions. *Dokl. Akad. Nauk. Russ. Acad. Sci.* **1963**, *148*, 1042–1045.
- 47. Constantine, P.G.; Diaz, P. Global sensitivity metrics from active subspaces. Reliab. Eng. Syst. Saf. 2017, 162, 1–13. [CrossRef]
- 48. Constantine, P.G.; Emory, M.; Larsson, J.; Iaccarino, G. Exploiting active subspaces to quantify uncertainty in the numerical simulation of the HyShot II scramjet. *J. Comput. Phys.* **2015**, *302*, 1–20. [CrossRef]
- 49. Constantine, P.G.; Dow, E.; Wang, Q. Active subspace methods in theory and practice: Applications to kriging surfaces. *SIAM J. Sci. Comput.* **2014**, *36*, A1500–A1524. [CrossRef]
- 50. Moraes, E.A.B.D.; KC, P.; Zayernouri, M. A Thermo-Electro-Mechanical Model for Long-Term Reliability of Aging Transmission Lines. *arXiv* 2024, arXiv:2406.18860.
- 51. National Centers for Environmental Information (NCEI), NOAA. Comparative Climatic Data (CCD). Available online: https://www.ncei.noaa.gov/products/land-based-station/comparative-climatic-data (accessed on 25 February 2024).
- 52. Service, N.W. National Weather Service. Available online: https://www.weather.gov/ (accessed on 25 February 2024).

- 53. Cengel, Y.A.; Ghajar, A.J.; Kanoglu, M. Heat and Mass Transfer: Fundamentals and Applications; McGraw Hill: New York, NY, USA, 2011.
- 54. Karoumi, R. Some modeling aspects in the nonlinear finite element analysis of cable supported bridges. *Comput. Struct.* **1999**, *71*, 397–412. [CrossRef]
- 55. Grigsby, L.L. *Electric Power Engineering Handbook*; CRC Press LLC: London, UK, 2006.
- 56. Reinoso, E.; Nino, M.; Berny, E.; Inzunza, I. Wind risk assessment of electric power lines due to hurricane hazard. *Nat. Hazards Rev.* 2020, *21*, 04020010. [CrossRef]
- 57. Holmes, J.D.; Paton, C.; Kerwin, R. Wind Loading of Structures; CRC Press: Boca Raton, FL, USA, 2007.
- Vasquez, W.A.; Jayaweera, D.; Játiva-Ibarra, J. End-of-life failure modelling of overhead lines considering loading and weather effects. In Proceedings of the 2019 IEEE International Conference on Power, Electrical, and Electronics and Industrial Applications (PEEIACON), Dhaka, Bangladesh, 29 November–1 December 2019.
- Cimini, C.A., Jr.; Fonseca, B.Q.A. Temperature profile of progressive damaged overhead electrical conductors. Int. J. Electr. Power Energy Syst. 2013, 49, 280–286. [CrossRef]
- 60. Abdi, H.; Williams, L.J. Principal component analysis. Wiley Interdiscip. Rev. Comput. Stat. 2010, 2, 433–459. [CrossRef]
- 61. Chevreuil, M.; Lebrun, R.; Nouy, A.; Rai, P. A least-squares method for sparse low rank approximation of multivariate functions. *SIAM/ASA J. Uncertain. Quantif.* **2015**, *3*, 897–921. [CrossRef]
- 62. Sobol, I. Sensitivity estimates for nonlinear mathematical models. Math. Model. Comput. Exp. 1993, 1, 407.
- 63. Saltelli, A.; Annoni, P.; Azzini, I.; Campolongo, F.; Ratto, M.; Tarantola, S. Variance based sensitivity analysis of model output. Design and estimator for the total sensitivity index. *Comput. Phys. Commun.* **2010**, *181*, 259–270. [CrossRef]
- 64. Machado, M.; Dos Santos, J. Reliability analysis of damaged beam spectral element with parameter uncertainties. *Shock Vib.* **2015**, 2015, 574846. [CrossRef]
- 65. Low, Y.M. A new distribution for fitting four moments and its applications to reliability analysis. *Struct. Saf.* **2013**, *42*, 12–25. [CrossRef]
- Dang, C.; Xu, J. Novel algorithm for reconstruction of a distribution by fitting its first-four statistical moments. *Appl. Math. Model.* 2019, 71, 505–524. [CrossRef]
- 67. Lasota, R.; Stocki, R.; Tauzowski, P.; Szolc, T. Polynomial chaos expansion method in estimating probability distribution of rotor-shaft dynamic responses. *Bull. Pol. Acad. Sci. Tech. Sci.* **2015**, *63*, 413–422. [CrossRef]
- 68. García, N.A.; Gómez, G.G.; Serna, C.N. Polinomial chaos expansion applied to limit state functions. J. Phys. Conf. Ser. 2021, 1981, 012010. [CrossRef]
- 69. He, J.; Gao, S.; Gong, J. A sparse grid stochastic collocation method for structural reliability analysis. *Struct. Saf.* **2014**, *51*, 29–34. [CrossRef]
- Winterstein, S.R.; MacKenzie, C.A. Extremes of nonlinear vibration: Comparing models based on moments, L-moments, and maximum entropy. J. Offshore Mech. Arct. Eng. 2013, 135, 021602. [CrossRef]

**Disclaimer/Publisher's Note:** The statements, opinions and data contained in all publications are solely those of the individual author(s) and contributor(s) and not of MDPI and/or the editor(s). MDPI and/or the editor(s) disclaim responsibility for any injury to people or property resulting from any ideas, methods, instructions or products referred to in the content.